Author’s response to reviews

Title: Factors influencing the use of reproductive health care services among married adolescent girls in Dang District, Nepal: a qualitative study.

Authors:

Binita Maharjan (binitamaharjan44@gmail.com)

Poonam Rishal (poonam.rishal1@gmail.com)

Joar Svanemyr (joars@cmi.no)

Version: 5 Date: 10 Apr 2019

Author’s response to reviews:

Dear editor

Thank you for your suggestions. We have anonymized the respondents in line with your recommendation. Concerning consent, we state that "For participants who were 17 years or under, and thus were considered children, permission was obtained from the husband or mother-in-law with the initial approval from the participant and informed assent form signed by the participant. The consent procedure used is in line with national guidelines and was approved by the ethics committee."

We do consider that we are aligned with the COREQ guidelines and most importantly concerning “report how participants were selected”, “describe the context in which the data were collected” and describe how data was collected.

The manuscript has been copy edited twice by different editors.

We hope this answers your concerns.

Best regards

Joar Svanemyr